Author's response to reviews

Title: Socioeconomic differentials in premature mortality in Rome: changes from 1990 to 2001

Authors:

Giulia Cesaroni (cesaroni@asplazio.it)
Nera Agabiti (agabiti@asplazio.it)
Francesco Forastiere (forastiere@asplazio.it)
Carla Ancona (ancona@asplazio.it)
Carlo A Perucci (perucci@asplazio.it)

Version: 5 Date: 2 November 2006

Author's response to reviews:

Editor
BioMed Central Public Health

(ID 1825600760100643)

The attached manuscript is submitted for publication after formatting changes.

Abstract word count: 197
Main text: 1919

Best regards
Giulia Cesaroni

Please address all the correspondence to:
Giulia Cesaroni, MSc
Department of Epidemiology, ASL RME
Via Santa Costanza 53,
00198 Roma, Italy
Phone: --39-06 83060462 (Direct)
Fax: --39-06 83060463
E-mail: cesaroni@asplazio.it

Authors and affiliations
Giulia Cesaroni, Nera Agabiti, Francesco Forastiere, Carla Ancona, and Carlo Alberto Perucci
Department of Epidemiology, Rome E Health Authority, Via di Santa Costanza 53, Rome 00198, Italy